Author’s response to reviews

Title: What motivates medical students to select medical studies: A systematic literature review

Authors:

Sonu Goel (sonugoel007@yahoo.co.in)

Federica Angeli (federica.angeli@maastrichtuniversity.nl)

Nonita Dhirar (nonita183@yahoo.com)

Neetu Singla (neetu.singla12@gmail.com)

Dirk Ruwaard (d.ruwaard@maastrichtuniversity.nl)

Version: 4 Date: 13 Dec 2017

Author’s response to reviews:

Dear Editors,

We thank the reviewers and your good-self for reviewing our paper. We are pleased that there is interest in the topic and are grateful for the constructive comments that could make the manuscript suitable for publication. We have addressed each of the reviewers’ comments point-by-point and have made corresponding changes in the manuscript.

Thanks and regards